Author’s response to reviews

Title: Evaluating the effectiveness of opportunistic eye screening model for people with Diabetes attending Diabetes clinic at Mnazi Mmoja hospital, Zanzibar

Authors:

Fatma J Omar (fatmakao@hotmail.com)
SheelaDevi Sethu (sheela@lvpei.org)
Padmaja Kumari Rani (rpk@lvpei.org)
Geng Ning (fatmakao@hotmail.com)
George Kabona (dr.georgekabona@yahoo.com)

Version: 2
Date: 28 January 2014

Author’s response to reviews: see over
To the Editor,

We would like to submit our manuscript “Evaluating the effectiveness of opportunistic eye screening model for people with Diabetes attending Diabetes clinic at Mnazi Mmoja hospital, Zanzibar for publication in your esteemed journal.

Diabetes and Diabetic retinopathy prevalence is showing an alarming rise in developing countries like Zanzibar. There is an urgent need to develop appropriate screening models to combat blindness due to Diabetic retinopathy. Our study is the first study to evaluate an opportunistic screening model for DR in Zanzibar.

We submitted revised version of manuscript with addition of key words, competing interests section and acknowledgments section.

Yours sincerely

Dr Padmaja Kumari Rani, M.S,FNB (Retina)
Srimati Kannuri Santhamma Centre for VitreoRetinal Diseases
Kallam Anji Reddy Campus,
L V Prasad Eye Institute
L V Prasad Marg, Hyderabad – 500 034
Ph: +91 040 30612607, Fax: +91 40 23548271
Email: rpk@lvpei.org Website: www.lvpei.org